Author's response to reviews

Title: Obesity prevention in child care: A review of U.S. state regulations

Authors:

Sara E Benjamin (sara_benjamin@harvardpilgrim.org)
Angie Cradock (acradock@hsph.harvard.edu)
Elizabeth Walker (emwalker@NEMOURS.ORG)
Meghan Slining (slining@email.unc.edu)
Matthew W Gillman (matthew_gillman@harvardpilgrim.org)

Version: 5 Date: 22 May 2008

Author's response to reviews: see over
**Title:** Obesity prevention in child care: A review of U.S. state regulations

Thank you for the opportunity to publish this manuscript in BioMed Central Public Health. We have made all requested formatting changes to the manuscript (please see requests below).

Thank you,
Sara Benjamin

--------------------------------------------
Authors, affiliations and email addresses - Please list these separately.

Qualifications - Please remove the authors' qualifications (e.g. PhD / MD) from the manuscript.

Affiliations - Please link each author to their corresponding affiliation with the use of superscript numbers. Please also remember to include the county in the affiliation details.

Email address - Please use the following format:
GSF: g.s.feder@qmul.ac.uk
HS: hsohal@yahoo.co.uk
SE: s.eldridge@qmul.ac.uk

Section headings - Please use sentence case in the section headings i.e. Abstract, not ABSTRACT.

INTRODUCTION - Please rename this section 'Background'.

The authors wish to declare that they have no competing interests - Please include this information under the heading 'Competing interests'.

Authors' contributions - Please state that all authors read and approved the final manuscript.

Table orientation- we prefer tables and figures to fit onto an portrait A4 page. If possible please re-orientate your table to the portrait orientation.